Author's response to reviews

Title: Exposure of the Swiss population to computed tomography

Authors:

Abbas Aroua (abbas@aroua.com)
Eleni-Theano Samara (elina.samara@hopitalvs.ch)
François O Bochud (francois.bochud@chuv.ch)
Reto Meuli (reto.meuli@chuv.ch)
Francis R Verdun (francis.verdun@chuv.ch)

Version: 2 Date: 26 December 2012

Author's response to reviews:

A section "Competing Interests" was added
A section "Authors' Contributions" was added
This study does not require an "Ethical Agreement".